Author's response to reviews

Title: Attachment-oriented psychological intervention for couples facing breast cancer: protocol of a randomised controlled trial

Authors:

Anne Nicolaisen (anicolaisen@health.sdu.dk)
Dorte G Hansen (dgilsaa@health.sdu.dk)
Mariët Hagedoorn (mariet.hagedoorn@med.umcg.nl)
Henrik E Flyger (henrik.flyger@regionh.dk)
Nina Rottmann (nrottmann@health.sdu.dk)
Per Nielsen (pernielsen74@gmail.com)
Katrine Søe (katrine.soee@rsyd.dk)
Anne E Pedersen (aepd@regionsjaelland.dk)
Christoffer Johansen (christof@CANCER.DK)

Version: 4 Date: 27 June 2014

Author's response to reviews: see over
Dear Editor

Thank you very much for your email. We are very pleased that you have accepted the transfer from BMC Cancer to BMC Psychology. We believe that our paper will be of interest to your readers being a study protocol on a large on-going randomized controlled trial with 199 breast cancer patients and their partners. The project has been approved by the Danish Data protection board (approval number: 2012-41-0392) and The Regional Scientific Ethics Committee for Southern Denmark (approval number S-20110100) and is registered at ClinicalTrials.gov (identifier NCT01368380).

We have responded to your two requests as best as we could. Please find the revised manuscript entitled “Attachment-oriented psychological intervention for couples facing breast cancer: protocol of a randomised controlled trial” submitted. The funding information has been transferred to and elaborated in the Acknowledgements section. The Competing Interests section has been properly completed.

This manuscript has not been published before, nor is it being considered for publication elsewhere. All authors have read and approved the manuscript. The authors have no financial or other conflicts of interest which could have biased our work. No other articles relating to this study protocol are yet published.

Thank you for your consideration of this manuscript. I look forward to hearing from you.

Best Regards

Anne Nicolaisen

MHS, PhD Student
Direct phone +45 6550 4258
E-mail anicolaisen@health.sdu.dk